Author’s response to reviews

Title: Gastroparesis is associated with oxytocin deficiency, oesophageal dysmotility with hyperCCKemia, and autonomic neuropathy with hypergastrinemia

Authors:

Julia Faraj (julia.faraj@med.lu.se)
Olle Melander (olle.melander@med.lu.se)
Linda Johansson (Linda.Johansson@his.se)
Kerstin Uvnäs-Moberg (Kerstin.Uvnas-Moberg@fyfa.ki.se)
Jens F Rehfeld (jens.f.rehfeld@rh.regionh.dk)
Bodil Ohlsson (bodil.ohlsson@med.lu.se)

Version: 4 Date: 16 February 2009

Author’s response to reviews:

Dear Editor,

1. I have stated that all authors have read and approved the final version.
2. I have now made all the figures to separate files.
3. The highlighted areas are not highlighted any more
4. All text is in black.
5. The abbreviation list is changed to a paragraph
6. The section conflict of interest is deleted.
7. The first author, Julia Faraj, has been married since the manuscript was submitted in september and changed her name to Borg. Her email adress is unchanged.

Sincerely Bodil Ohlsson